Author's response to reviews

Title: Study of the therapeutic effects of an advanced hippotherapy simulator in children with cerebral palsy: a randomised controlled trial.

Authors:

Pablo Herrero (pherrero@usj.es)
Angel Asensio (aasensio@unizar.es)
Elena García (elenaga@aidimo.org)
Alvaro Marco (amarco@unizar.es)
Bárbara Olivan (bolivan@usj.es)
Alejandro Ibarz (aibarz@unizar.es)
Eva M Gómez-Trullén (evagomez@unizar.es)
Roberto Casas (rcasas@unizar.es)

Version: 5 Date: 12 April 2010

Author's response to reviews:

I have deleted a sentence which showed a link to table 1 because it was not correct.

In the author’s contribution, as it has been the last part submitted, our English reviewer has made some writing changes